Author’s response to reviews

Title: Fit for Fight – Self-Reported Health in Military Women: a cross-sectional study.

Authors:

Elin Fadum (fadumelin@gmail.com)
Leif Strand (Leif.Age.Strand@krefregisteret.no)
Monica Martinussen (monica.martinussen@uit.no)
Laila Breidvik (lbreidvik@mil.no)
Nina Isaksen (ninaisaksen@gmail.com)
Einar Borud (einar.borud@uit.no)

Version: 2 Date: 09 Sep 2019

Author’s response to reviews:

1. The participants gave their written consent to participate in the study. This is now clarified at page 20 line 446.

2. We have revised the authors contributions according to your instructions and declare that they have contributed in line with the ICMJE guidelines for authorship.

3. We have added a list of additional files after the references, and secured that the additional files are cited in the manuscript.

4. We have placed all the figure titles (Figure 1-4) under the heading “Figure headings” after the references.